Author's response to reviews

Title: Collection of population-based cancer staging information in Western Australia - a feasibility study

Authors:

Tim Threlfall (tim.threlfall@health.wa.gov.au)
Jana Wittorff (janaw@ferriscan.com)
Noy Boutdara (padabphet@hotmail.com)
Jane Heyworth (jane.heyworth@uwa.edu.au)
Paul Katris (pkatris@cancerwa.asn.au)
Harry Sheiner (wacog@cancerwa.asn.au)
Lin Fritschi (lfritschi@qldcancer.com.au)

Version: 2 Date: 16 August 2005

Author's response to reviews:

Revision as requested by Gemma Burns
Editorial Production Assistant